Author's response to reviews

Title: Early pubertal onset and its relationship with sexual risk taking, substance use and anti-social behaviour: a preliminary cross-sectional study

Authors:

Jennifer Downing (j.downing1@ljmu.ac.uk)
Mark A Bellis (m.a.bellis@ljmu.ac.uk)

Version: 6 Date: 26 November 2009

Author's response to reviews:

Dear editor,

I have made very slight changes to two numbers in tables 2 and 4 where they had been automatically rounded instead of being xx.0

I have also slightly changed the spacing in tables 4 and 5 as it was difficult to see which variables related to each section without the lines.

Many thanks

Kind regards

Jennifer Downing